Author's response to reviews

Title: Gut Microbiota, Probiotics and Diabetes

Authors:

Aline C Gomes (aline2nut@hotmail.com)
Allain A Bueno (a.bueno@worc.ac.uk)
Rávila G M Souza de (ravilagraziany@gmail.com)
João F Mota (jfemota@gmail.com)

Version: 3 Date: 30 May 2014

Author's response to reviews: see over
To
Ms. Rovien Reyes
The Nutrition Journal Editorial Team

30th May 2014

Dear Ms Reyes:
Ref: Manuscript 2097875960112759 GUT MICROBIOTA, PROBIOTICS AND DIABETES

Thank you for your letter dated 8th May 2014. We are grateful to the reviewers for their helpful comments. As suggested, minor essential revisions have been made.

Please find attached the amended version of our manuscript and a reply to the reviewers.

Thank you for your consideration and we look forward to hear from you.

Yours sincerely,
Dr Mota, on behalf of all authors.